Author's response to reviews

Title: Antifungal activities of essential oil extract of Carpobrotus edulis (L.) Bolus

Authors:

Beauty E Omoruyi Dr (sayhi2sa@yahoo.com)
Anthony A.J Afolayan Prof (aafolayan@ufh.ac.za)

Version: 2 Date: 28 October 2013

Author's response to reviews:

Dear editor,

I have done the following corrections in my article:

1. The name of third author is now G. Bradley as against Bradley G.
2. The email address of third author is now gbradley@ufh.ac.za instead of GBradley@ufh.ac.za

Please can you try this order of arrangements, because at the moment we don't know what else to do. Thanks for the understanding and anticipated cooperation.

Kind regards,

Beauty E.

The correct list of the authors have been uploaded